Author’s response to reviews

**Title:** Daily routine vs on-demand chest radiograph policy in adult ICU patients- Clinicians' perspective

**Authors:**
Abdullah AlShahrani (alshahrani010@gmail.com)
khaled Al-Surimi (kalsurimi@gmail.com)

**Version: 1 Date: 19 Mar 2017**

**Author’s response to reviews:**

Editor's comments:

1. Please clarify in your Ethics statement under the Declarations section that each site has provided ethical approval for this study.

   - We agree with the editor and the following statement is included "Each site has provided ethical approval for this study and all the participants provided written informed consent".

2. Thank you for including a data availability statement. This statement should detail the location of the raw data underlying the conclusions made in this study. We note that only summarized data is presented in this manuscript, so we would like to ask that you amend this statement. If there are any ethics restrictions preventing the sharing of the raw data, please state this. If you do not wish to share the data, please include a reason why it will not be shared in this section.

   - The following statement was included under availability of data section;

   "The datasets used and/or analyzed during the current study available from the corresponding author on reasonable request."